Author's response to reviews

Title: Fatigue Intervention by Nurses Trial - The FINE Trial. A randomised controlled trial of nurse led self-help treatment for patients in primary care with chronic fatigue syndrome: study protocol. [ISRCTN 74156610]

Authors:

Alison J Wearden (alison.wearden@manchester.ac.uk)
Lisa Riste (lisa.riste@manchester.ac.uk)
Christopher Dowrick (cfd@liverpool.ac.uk)
Carolyn Chew-Graham (carolyn.chew-graham@manchester.ac.uk)
Richard P Bentall (richard.bentall@manchester.ac.uk)
Richard K Morriss (richard.morriss@nottingham.ac.uk)
Sarah Peters (speters@liv.ac.uk)
Graham Dunn (graham.dunn@manchester.ac.uk)
Gerry Richardson (gar2@york.ac.uk)
Karina Lovell (karina.lovell@manchester.ac.uk)
Pauline Powell (pauline.powell@rlbuht.nhs.uk)

Version: 4 Date: 4 April 2006

Author's response to reviews:

I have accepted all the changes you made to the manuscript.
I have not made any amendments.